Author's response to reviews

Title: The Centella asiatica Juice effects on DNA damage, Apoptosis and Gene Expression in Hepatocellular carcinoma (HCC)

Authors:

Faridah Hussin (dka_as@yahoo.com)
Sima Ataollahi Eshkoor (simaataolah@yahoo.com)
Asmah Rahmat (satirann@yahoo.com)
Fauziah Othman (s1_ataolah@yahoo.com)
Abdah Akim (s1_ataolah@yahoo.com)

Version: 8 Date: 17 October 2013

Author's response to reviews: see over
Dear Editor,

I do appreciate your valued suggestions of your second round of reviewing.

Hope the corrections would be satisfactory.

1. Voucher specimen of material already was mentioned
2. Figure legends were removed
3. English writing was checked and edited

Sincerely yours

Asmah Rahmat